Author's response to reviews

Title: An analysis of trends and determinants of health insurance and healthcare utilisation in the Russian population between 2000 and 2004: the 'inverse care law' in action

Authors:

Francesca Perlman (francesca.perlman@lshtm.ac.uk)
Dina Balabanova (dina.balabanova@lshtm.ac.uk)
Martin McKee (martin.mckee@lshtm.ac.uk)

Version: 3 Date: 7 April 2009

Author's response to reviews:

Formatting changes requested (your email dated 5th April 2009)

References 1 and 6 are now published - I have added the complete citations
Table vertical elements - these have all been removed
Tables 3 and 4 - these have both been reformatted to fit onto an A4 sheet
Reference order - these should all be in order - I used the Endnote programme with the BMC HSR template. There were 2 references which had lost their formatting - I have reformatted them using the programme.
Country - this has been added to the front page to the affiliations